Author’s response to reviews

Title: Optimizing search strategies to identify randomized controlled trials in MEDLINE

Authors:

Li Zhang (li.zhang@usask.ca)
Isola Ajiferuke (iajiferu@uwo.ca)
Margaret Sampson (msampson@cheo.on.ca)

Version: 6 Date: 8 May 2006

Author’s response to reviews:

Dear Editor,

We are pleased to know that the manuscript is accepted in principle, and thanks for giving the formatting instructions. We have changed the format of the manuscript to be consistent with the journal requirements.

Sincerely,

Li Zhang